Author's response to reviews

Title: A prospective observational study to assess the diagnostic accuracy of clinical decision rules for children presenting to emergency departments after head injuries (protocol): The Australasian Paediatric Head Injury Rules Study (APHIRST).

Authors:

Franz E Babl (franz.babl@rch.org.au)
Mark Lyttle (mdlyttle77@gmail.com)
Meredith Borland (Meredith.borland@health.wa.gov.au)
Natalie Phillips (Natalie.phillips@merton-oxford.com)
Amit Kochar (Amit.kochar@health.sa.gov.au)
Stuart R Dalziel (sdalziel@adhb.govt.nz)
Sarah Dalton (Sarah.dalton@health.nsw.gov.au)
John Cheek (John.cheek@monashhealth.org)
Jeremy Furyk (Jeremy.furyk@health.qld.gov.au)
Yuri Gilhortra (Yuri.gilhortra@mater.org.au)
Jocelyn Neutze (Jocelyn.neutze@middlemore.co.nz)
Silvia Bressan (Silviabress@gmail.com)
Ed Oakley (Ed.oakley@rch.org.au)
Louise Crowe (Louise.crowe@mcri.edu.au)
Susan Donath (Susan.donath@mcri.edu.au)
Kim Jachno (Kim.jachno@mcri.edu.au)
Brenton Ward (Brenton.ward@mcri.edu.au)
Amanda Williams (Amanda.williams@rch.org.au)

Version: 2
Date: 8 May 2014

Author's response to reviews:

Dear Editors,

We have revised the manuscript with your suggested changes.
1. We have included the trial registration date in the abstract
2. We have included a discussion heading and section in the manuscript
3. We have included line numbering in the manuscript

Please do not hesitate to contact me if you have any more queries.

Best regards

Franz E Babl